Author's response to reviews

Title: Lifestyle interventions for overweight and obese pregnant women to improve pregnancy outcome: systematic review and meta-analysis

Authors:

Eugene Oteng-Ntim (eugene.oteng-ntim@gstt.nhs.uk)
Rajesh Varma (rajesh.varma@gstt.nhs.uk)
Helen Crocker (helen.crocker@ucl.ac.uk)
Lucilla Poston (lucilla.poston@kcl.ac.uk)
Pat Doyle (pat.doyle@lshtm.ac.uk)

Version: 4 Date: 26 March 2012

Author's response to reviews:

1. Authors’ contributions: please confirm that all authors approved the final version of the manuscript in this section. YES. ALL AUTHORS APPROVE OF FINAL VERSION

2. Please clarify in writing whether all figures were made exclusively for this manuscript, or whether copyright permission was obtained. ALL FIGURES WERE MADE EXCLUSIVELY FOR THIS MANUSCRIPT.